Author’s response to reviews

Title: Role of external facilitation in implementation of research findings: a qualitative evaluation of facilitation experiences in the Veterans Health Administration

Authors:

Cheryl B Stetler (Cheryl.Stetler@comcast.net)
Marcia W Legro (mlegro@earthlink.net)
Joanne Rycroft-Malone (j.rycroft-malone@bangor.ac.uk)
Candice Bowman (Candice.Bowman@va.gov)
Geoffrey Curran (CURRAN.GEOFFREYM@va.gov)
Marylou Guihan (Marylou.guihan@va.gov)
Hildi Hagedorn (Hildi.Hagedorn@va.gov)
Sandra Pineros (Sandra.Pineros@va.gov)
Carolyn M Wallace (Carolyn.Wallace1@va.gov)

Version: 5 Date: 8 October 2006

Author’s response to reviews:

Reference call-outs and underlining corrected.